

The fall and rise of the global climate model

Article

Published Version

Creative Commons: Attribution 4.0 (CC-BY)

Open Access

Muelmenstaedt, J. and Wilcox, L. J. ORCID: https://orcid.org/0000-0001-5691-1493 (2021) The fall and rise of the global climate model. Journal of Advances in Modeling Earth Systems, 13 (9). ISSN 1942-2466 doi: https://doi.org/10.1029/2021MS002781 Available at http://centaur.reading.ac.uk/100305/

It is advisable to refer to the publisher's version if you intend to cite from the work. See Guidance on citing.

Published version at: https://agupubs.onlinelibrary.wiley.com/doi/10.1029/2021MS002781
To link to this article DOI: http://dx.doi.org/10.1029/2021MS002781

Publisher: American Geophysical Union

All outputs in CentAUR are protected by Intellectual Property Rights law, including copyright law. Copyright and IPR is retained by the creators or other copyright holders. Terms and conditions for use of this material are defined in the End User Agreement.

www.reading.ac.uk/centaur

CentAUR

Central Archive at the University of Reading



Reading's research outputs online





JAMES | Journal of Advances in Modeling Earth Systems®



COMMENTARY

10.1029/2021MS002781

Key Points:

- Global models are deficient in how they represent the aerosol influence on clouds and regional circulation patterns
- Nevertheless, global models are essential for reliable regional climate projections
- Recent advances make it more likely that future global models will be greatly improved

Correspondence to:

J. Mülmenstädt, johannes.muelmenstaedt@pnnl.gov

Citation:

Mülmenstädt, J., & Wilcox, L. J. (2021). The fall and rise of the global climate model. *Journal of Advances in Modeling Earth Systems*, 13, e2021MS002781. https://doi.org/10.1029/2021MS002781

Received 17 AUG 2021 Accepted 25 AUG 2021

The Fall and Rise of the Global Climate Model

Johannes Mülmenstädt¹ D and Laura J. Wilcox^{2,3} D

¹Atmospheric Sciences & Global Change Division, Pacific Northwest National Laboratory, Richland, WA, USA, ²National Centre for Atmospheric Science, University of Reading, Reading, UK, ³Department of Meteorology, University of Reading, Reading, UK

Abstract Global models are an essential tool for climate projections, but conventional coarseresolution atmospheric general circulation models suffer from errors both in their parameterized cloud physics and in their representation of climatically important circulation features. A notable recent study by Terai et al. (2020, https://doi.org/10.1029/2020ms002274) documents a global model capable of reproducing the regime-based effect of aerosols on cloud liquid water path expected from observational evidence. This may represent a significant advance in cloud process fidelity in global models. Such models can be expected to give a better estimate of the effective radiative forcing of the climate. If this advance in cloud process representation can be matched by advances in the representation of circulation features such as monsoons, then such models may also be able to navigate the complex tangle between spatially heterogeneous aerosol-cloud interactions and regional circulation patterns. This tight link between aerosol and circulation results in anthropogenic perturbations of climate variables of societal importance, such as regional rainfall distributions. Upcoming global models with km-scale resolution may improve the regional circulation and be able to take advantage of the Terai et al. (2020, https://doi. org/10.1029/2020ms002274) improvement in cloud physics. If so, an era of significantly improved regional climate projection capabilities may soon dawn. If not, then the improvement in cloud physics might spur intensified efforts on problems in model dynamics. Either way, based on the rapid changes in aerosol emissions in the near future, learning to make reliable projections based on biased models is a skill that will not go out of style.

Plain Language Summary Human activity releases particles into the atmosphere. Some of these particles are small enough to remain suspended in air, forming an "aerosol." Aerosols interact with clouds to make them brighter or dimmer. Clouds are very good sunlight reflectors, so changing their properties even slightly results in large changes to how much sunlight is absorbed by the Earth. Over time, this changes the rate at which the climate warms. It also affects the regional circulation, that is, recurring weather patterns at the regional scale. This is especially important to society when it affects regions that experience intense rainfall, such as the summer monsoon. If circulation patterns change, locations that used to experience intense rain may receive far less rain than they are used to, and vice versa. To predict how human activity will change regional rainfall in the future requires using models. Unfortunately, the models that are currently available struggle with both the interactions between aerosols and clouds and with some regional circulation features, so their regional predictions are not as reliable as society needs them to be. A recent study describes a model that appears to be much better at aerosol-cloud interactions than previous global models, potentially solving half the problem. Many groups are currently at work on a new generation of models that may be better at regional circulation, which would solve the other half, although it may be good to temper any optimism until this has actually been demonstrated. If these two advances can be combined within the same model, then that model will be able to provide a much more reliable estimate of future regional climate. But model development is hard and slow work, while aerosol emissions are changing rapidly, so it is also important to find ways to extract useful information from imperfect models.

© 2021 Battelle Memorial Institute. This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

The atmosphere is a tangle of tangles: a web of competing and compensating cloud processes, interwoven with the atmospheric circulation, which itself is threaded across vastly different spatiotemporal scales. Anthropogenic aerosol perturbations subtly shift this web of interactions. This introduces imbalances in the planet's energy budget on the order of a percent of the unperturbed radiative energy fluxes and moves circulation features, such as the ITCZ and monsoons, with small changes at continental scales having large

MÜLMENSTÄDT AND WILCOX



impacts at local scales. For models to be able to project these changes, they need to represent the tangle of diabatic processes, the tangle of atmospheric circulation, and the tangle of those two tangles, each at the appropriate level of complexity. This level of complexity is the minimal degree of model elaboration required to reproduce the emergent properties of the system. It varies with resolution, and finding it has been elusive. One approach is the "superparameterization" multiscale modeling framework (Randall et al., 2003), which conjoins a general circulation model (GCM) to represent large scales with a higher-resolution model running in each GCM column to begin probing cloud process scales. Initially, this approach was devised for deep convection, with the convection handled by a km-scale cloud-resolving model (CRM) (Grabowski & Smolarkiewicz, 1999; Khairoutdinov & Randall, 2001). Parishani et al. (2017) extended the method to shallow clouds by increasing the resolution of the CRM, enabling this "ultraparameterized" model to estimate low-cloud feedbacks (Parishani et al., 2018). Terai et al. (2020) describe the latest development, which endows the Parishani et al. (2017) ultraparameterized model with a representation of aerosol-cloud interactions. A novel result in this work is the model's qualitative agreement with one of the key aspects of observed regime dependence in aerosol-cloud interactions, namely that increasing aerosol concentration increases the liquid water path of precipitating warm clouds but decreases the liquid water path of nonprecipitating warm clouds. This behavior has not been reported in a global model before, potentially because of a structural bias in the model physics, which explicitly represents a mechanism for aerosols thickening raining clouds but not for thinning nonraining clouds. If the behavior of the ultraparameterized model is indeed due to the closing of this gap in the model physics, then Terai et al. (2020) have taken an important step toward identifying the appropriate level of complexity for modeling the climate impacts of aerosols, including their role in regional climate projections.

How do Terai et al. (2020) improve the representation of aerosol-cloud interactions in global models? The perturbation to the Earth's energy budget, on time scales faster than the warming of the ocean surface, is often referred to as the effective radiative forcing (ERF) by aerosol-cloud interactions (Boucher et al., 2014; Myhre et al., 2014). This is the sum of the near-instantaneous effect of more numerous, smaller droplets when aerosol concentrations are increased (Twomey, 1977), termed the radiative forcing (RF); and the slower (hours to days) responses of the clouds, termed adjustments, that follow the change in droplet number. Under the right set of conditions, the smaller cloud droplets in higher-aerosol conditions are slower to form precipitation (Squires, 1958), which increases average cloud amount (Albrecht, 1989) and thickness (Pincus & Baker, 1994). But they also evaporate more easily, which can beget a turbulent feedback that leads to even more evaporation, reducing average cloudiness (Ackerman et al., 2004; Bretherton et al., 2007). Paradoxically, what happens at the raindrop scale (micrometers to millimeters) is easier to include in the mechanics of a coarse-resolution model than what happens at the turbulent scale (meters to hundreds of meters). This results in an asymmetry in the treatment of these opposite-sign effects (Michibata et al., 2016; Mülmenstädt & Feingold, 2018; Salzmann et al., 2010; C. Zhou & Penner, 2017), with a big part of the physics simply missing or potentially overwhelmed by artificial mixing across a poorly resolved cloud-top inversion at the model scale. In the observed atmosphere, the liquid water path responds weakly to aerosols (Malavelle et al., 2017; Toll et al., 2019) or perhaps slightly decreases (Gryspeerdt et al., 2019), consistent with an enhanced evaporation effect sufficiently large to cancel or perhaps exceed the precipitation suppression effect that is explicitly represented in most models. Thus, while RF is far from well constrained (Quaas et al., 2020), the adjustments are where traditional cloud parameterizations used in GCM-scale models appear to go most wrong. In Terai et al. (2020), the CRM running within each GCM column has a resolution approaching large eddy simulations (LESs) suitable for boundary-layer clouds. LES at this resolution is known to be able to represent entrainment, albeit with quantitative uncertainty, through a combination of the resolved dynamics and subgrid turbulence parameterization. Our interpretation is that the model used by Terai et al. (2020) has been the first to hit the sweet spot between global coverage, resolution, and model physics required to understand the balance between precipitation suppression and evaporation enhancement.

There are two major problems with approaches that do not include global modeling. The first is that models that come closer to resolving the relevant cloud processes are too expensive to run across the full spectrum of conditions that real clouds encounter, a problem naturally solved by ergodicity in a climatology-length global model run. Second, the atmospheric circulation—at scales larger than are accessible to LES runs—and the spatial pattern of aerosol forcing mutually influence each other. Anthropogenic aerosol changes have been the dominant driver of trends in the ITCZ and global monsoons since the mid-twentieth century

MÜLMENSTÄDT AND WILCOX 2 of 7



(Lau & Kim, 2010; Polson et al., 2014; Ridley et al., 2015; Song et al., 2014), which in turn influences cloud distribution and aerosol distribution and deposition. The change in regional rainfall that accompanies the changes in these circulation features is one of the aspects of climate change that society most urgently requires us to project accurately. Therefore, both for global and especially for regional climate projections, global models are the only tool that allows us to explore the tight links between large-scale circulation and strongly localized aerosol sources.

As a proof of concept, the more realistic cloud physics behavior in the Terai et al. (2020) model is highly encouraging, but some drastic trade-offs were needed to get an ERF estimate. The host GCM is too coarse to resolve regional circulations crucial to aerosol–circulation interactions, likely a step backwards in representing regional circulations accurately given moderate improvements in the Asian summer monsoon (Goswami & Goswami, 2017; Johnson et al., 2016; Ramu et al., 2016) and substantial improvements in the ITCZ (Song & Zhang, 2020) with increasing GCM resolution. The embedded model is too coarse to resolve the turbulent feedback leading to enhanced evaporation, so its quantitative representation of boundary layer entrainment is likely uncertain. The model runs are also too short to sample internal variability in the atmospheric circulation, so distinguishing between forced response and internal variability is not possible. Thus, this is not yet the modeling system the community needs in order to make reliable climate projections. The global km-resolution models currently being developed by a number of groups may offer a pragmatic solution. They have the potential of outright solving enough of the cloud physics and the circulation problem, and, if so, they inherently solve the problem of the interaction between clouds and circulation. Barring that, if they lead to significant progress on only one of these problems, that may sharpen the community's focus on solving the other.

On the cloud physics side, the enhanced evaporation mechanism, which occurs at process scales on the order of meters (Mellado, 2017), is still far beyond the resolved scale of km-scale models. But it is also far beyond the resolved scale of the 250 m CRM in Terai et al. (2020). While the parameterized turbulence and cloud physics will have to do a lot of heavy lifting at the km scale, their task is less gargantuan than at the traditional GCM scale. Approaches exist for representing the cloud-scale turbulence in km-scale models, by eddy-diffusivity-mass-flux (EDMF) schemes (e.g., Siebesma et al., 2007; Tan et al., 2018) or higher-order turbulence closures (e.g., Bogenschutz & Krueger, 2013; Cheng & Xu, 2011; M. Wang et al., 2015), and the enhanced evaporation mechanism can be included in the parameterized physics (Guo et al., 2011). Fundamentally, the struggle is how to represent the limited mixing between the boundary layer and the free troposphere, which in the real stratocumulus-topped boundary layer are separated by a sharp inversion. In models with limited resolution, numerical diffusion leads to artificial mixing, perhaps great enough to drown out any parameterized microphysics-turbulence interaction that would represent the enhanced evaporation effect. For a given combination of model resolution and physics, the strength of the artificial mixing could give an indication of how parameterizable the aerosol-enhanced evaporation effect is in that model (Karset et al., 2020; Sato et al., 2018). Indications are that Terai et al. (2020) have found a combination of resolution and physics that mitigates the numerical diffusion problem; quantifying the numerical diffusion in other combinations of resolution and turbulence scheme (be it ultraparameterization, higher-order closures, EDMF, or any other) could identify other viable combinations. The advance that such models represent is trading a structural problem—a missing, or perhaps drowned out, physical mechanism—for a parametric one. While parametric uncertainty is a tough problem (Lee et al., 2016; Regayre et al., 2018; Tsushima et al., 2020), we have a hunch (Mülmenstädt et al., 2020, 2021) that the right types of observables in this case, perhaps the regime-dependent liquid water path susceptibilities of Gryspeerdt et al. (2019) make it more tractable than it currently appears.

On the circulation side, km-scale models' explicit, rather than parameterized, deep convection may improve their regional-scale circulation features (Oouchi et al., 2009), for example by better capturing the interaction between convection and circulation that is an essential feature of the monsoon initiation (Marsham et al., 2013; Willetts et al., 2017). However, regional biases can have remote origins (C. Wang et al., 2014), so long-standing biases in the global circulation, such as the position and structure of the eddy-driven jet. also need to be tackled. These may also benefit from the improved representation of orography that goes alongside increased horizontal resolution (Berckmans et al., 2013). As more groups begin performing longer model runs at higher resolution, we look forward to learning which circulation problems they can

MÜLMENSTÄDT AND WILCOX 3 of 7



mitigate, and which circulation problems will be thrown into starker relief by potential improvements in the cloud physics.

Models have an essential role to play in filling gaps in our understanding of the climate effects of aerosol-cloud interactions. Theories of precipitation suppression as a microphysical effect and enhanced evaporation as an interaction between microphysics and turbulence have existed for quite some time, as have observations of cloud and aerosol state. However, they are not sufficient to paint a global picture. At least in part, this is because theories of phenomena at one spatiotemporal scale in a complex system have limited predictive power on other scales. Models can synthesize our process knowledge across scales—from the cloud scale to the global circulation. Global models can sample the full range of meteorological variability and covariability in a way that more localized cloud process scale modeling cannot. Models based on physical process understanding, unlike empirical relationships in the present day, can provide accurate projections into a future where these empirical relationships may no longer hold. Importantly, they can also establish causation, which observations alone cannot. Global satellite datasets show what can be interpreted as the respective fingerprints of precipitation suppression and enhanced evaporation by aerosols in precipitating and nonprecipitating clouds (Gryspeerdt et al., 2019). The total observed effect of aerosols on liquid water path appears to be small (Malavelle et al., 2017; Toll et al., 2019). The observational evidence is consistent with opposing mechanisms that come close to canceling, but it cannot prove that this is the case. Confounding by meteorological covariability or by other processes cannot be disentangled from the processes hypothesized to cause the observed effect, limiting what observations alone can tell us. Model experiments where processes can be disabled or decoupled from each other, and where aerosol properties can be varied independently of meteorology, can give us a clear answer, provided that the model represents the processes under investigation. If causation can be demonstrated, global models can go further and translate the observed relationships between liquid water path and aerosols into parameters that are left unspecified by theory, namely the strength of precipitation suppression and evaporation enhancement. One of the rays of hope emanating from Terai et al. (2020) is that global models may soon attain the required level of physical completeness where they can play translator between observations and theory. The situation is similar in the theory and observation of regional circulations. There is a plethora of theoretically established mechanisms contributing to monsoons, the seasonal cycle of insolation, differential heating of land and ocean, and orography among them. But theory and the observational record alone do not translate into sufficient quantitative understanding of these mechanisms, their interactions with each other, and their interactions with external influences to yield the reliable regional climate projections that we need. In sufficiently complete models, process denial experiments could again translate between theory and observations.

Improving physics parameterizations is likely to expose remaining errors in models, such as those in largescale atmospheric circulation patterns. While global models are essential for establishing causation, biases in their representation of the global circulation can limit their simulation of the response to forcing, in addition to causing uncertainty in aerosol ERF. This is particularly pressing for near-term (20-40 years) projections, where the combination of scenario, radiative, and dynamical uncertainty makes anthropogenic aerosol a key factor in the spread in near-term projections of regional climate (Wilcox et al., 2020). Fortunately, the combination of theory, observations, and modeling may help to overcome deficiencies in any one. For example, recent decades have seen a weaker East Asian summer monsoon circulation, leading to flooding in southeast China and drought in the north. Global climate models capture this weakened circulation (Song et al., 2014; Wilcox et al., 2015), and indicate that it is driven by increases in anthropogenic aerosols (Li et al., 2015; Song et al., 2014). However, systematic errors in their representation of the monsoon circulation mean that the associated precipitation change is not correctly captured. In examples like this, where we can learn the mechanism underlying a change from models, there is the potential to combine models with observations to apply physically informed bias corrections to improve our confidence in projections, despite the known biases in the models (Zhou et al., 2020). Such approaches may be a solution when model developments cannot keep pace with society's need for projections.

For over a decade, the climatic effect of anthropogenic aerosol has appeared an intractable problem. Largely separate bodies of literature have attempted to tackle the uncertainties in global-mean ERF and regional climate responses. But these are strongly interwoven problems, and a satisfactory solution—one that permits reliable projections of regional climate change—requires a more unified treatment. This reunification

MÜLMENSTÄDT AND WILCOX 4 of 7



of mechanisms would be aided by more reliable global models, which are, at least in principle, capable of synthesizing the cross-scale interactions from cloud processes to the planetary circulation that have so far stymied every tool at our disposal: theoretical understanding, empirical observations, process-scale modeling, and global modeling. The results in Terai et al. (2020) make us hopeful that cloud physics is not an insurmountable problem in km-scale global models. There is also reason to be optimistic that these km-scale models, by better capturing the interplay between circulation and convection, may resolve some (but by no means all) longstanding circulation biases. We take this as evidence that the arc of our ability to model the climate response to aerosol is bending in the right direction, toward reliable projections of regional climate change, toward helping us test hypotheses built to explain observations, and toward giving global context to the process-scale understanding derived from LES of a limited set of conditions. It is a slowly bending arc, however, fundamentally limited by the fact that a perfect model is neither possible (Oreskes et al., 1994) nor perhaps desirable (Emanuel, 2020). There is no similar limitation on the speed with which society can continue to change its impact on the climate, so continuing to learn how to interpret the messages of biased models, alongside improving the models, remains an essential skill.

Acknowledgments

The authors are grateful to Robert Pincus, Ruby Leung, Bjorn Stevens, Po-Lun Ma, and Christina Sackmann for stimulating discussions and clarifying comments. J. Mülmenstädt was supported by the U.S. Department of Energy (DOE), Office of Science, Office of Biological and Environmental Research, Earth System Model Development program's "EAGLES" project (74358). L.J. Wilcox was supported by the Natural Environment Research Council (NERC: grant NE/S004890/1. EMERGENCE). The Pacific Northwest National Laboratory is operated for the U.S. DOE by Battelle Memorial Institute under contract DE-AC05-76RL01830.

References

- Ackerman, A., Kirkpatrick, M., Stevens, D., & Toon, O. (2004). The impact of humidity above stratiform clouds on indirect aerosol climate forcing. *Nature*, 432(7020), 1014–1017. https://doi.org/10.1038/nature03174
- Albrecht, B. A. (1989). Aerosols, cloud microphysics, and fractional cloudiness. Science, 245(4923), 1227–1230. https://doi.org/10.1126/science.245.4923.1227
- Berckmans, J., Woollings, T., Demory, M.-E., Vidale, P.-L., & Roberts, M. (2013). Atmospheric blocking in a high resolution climate model: Influences of mean state, orography and eddy forcing. *Atmospheric Science Letters*, 14(1), 34–40. https://doi.org/10.1002/asl2.412
- Bogenschutz, P. A., & Krueger, S. K. (2013). A simplified pdf parameterization of subgrid-scale clouds and turbulence for cloud-resolving models. *Journal of Advances in Modeling Earth Systems*, 5(2), 195–211. https://doi.org/10.1002/jame.20018
- Boucher, O., Randall, D., Artaxo, P., Bretherton, C., Feingold, G., Forster, P., & Zhang, X. (2014). Clouds and aerosols. In Climate change 2013: The physical science basis. Contribution of working group I to the fifth assessment report of the intergovernmental panel on climate change (pp. 571–658). Cambridge, United Kingdom and New York, NY: Cambridge University Press. https://doi.org/10.1017/CBO9781107415324.016
- Bretherton, C. S., Blossey, P. N., & Uchida, J. (2007). Cloud droplet sedimentation, entrainment efficiency, and subtropical stratocumulus albedo. *Geophysical Research Letters*, 34(3). https://doi.org/10.1029/2006GL027648
- Cheng, A., & Xu, K.-M. (2011). Improved low-cloud simulation from a multiscale modeling framework with a third-order turbulence closure in its cloud-resolving model component. *Journal of Geophysical Research*, 116, D14101. https://doi.org/10.1029/2010JD015362
- Emanuel, K. (2020). The relevance of theory for contemporary research in atmospheres, oceans, and climate. AGU Advances, 1(2), e2019AV000129. https://doi.org/10.1029/2019AV000129
- Goswami, B. B., & Goswami, B. N. (2017). A road map for improving dry-bias in simulating the South Asian monsoon precipitation by climate models. Climate Dynamics, 49(5–6), 2025–2034. https://doi.org/10.1007/s00382-016-3439-2
- Grabowski, W. W., & Smolarkiewicz, P. K. (1999). CRCP: A cloud resolving convection parameterization for modeling the tropical convecting atmosphere. *Physica D*, 133(1–4), 171–178. https://doi.org/10.1016/S0167-2789(99)00104-9
- Gryspeerdt, E., Goren, T., Sourdeval, O., Quaas, J., Mülmenstädt, J., Dipu, S., et al. (2019). Constraining the aerosol influence on cloud liquid water path. *Atmospheric Chemistry and Physics*, 19(8), 5331–5347. https://doi.org/10.5194/acp-19-5331-2019
- Guo, H., Golaz, J.-C., & Donner, L. J. (2011). Aerosol effects on stratocumulus water paths in a pdf-based parameterization. *Geophysical Research Letters*, 38, L17808. https://doi.org/10.1029/2011GL048611
- Johnson, S. J., Levine, R. C., Turner, A. G., Martin, G. M., Woolnough, S. J., Schiemann, R., et al. (2016). The resolution sensitivity of the South Asian monsoon and indo-pacific in a global 0.35° AGCM. Climate Dynamics, 46(3–4), 807–831. https://doi.org/10.1007/s00382-015-2614-1
- Karset, I. H. H., Gettelman, A., Storelvmo, T., Alterskjaer, K., & Berntsen, T. K. (2020). Exploring impacts of size-dependent evaporation and entrainment in a global model. *Journal of Geophysical Research*, 125(4), e2019JD031817. https://doi.org/10.1029/2019JD031817
- Khairoutdinov, M. F., & Randall, D. A. (2001). A cloud resolving model as a cloud parameterization in the NCAR community climate system model: Preliminary results. *Geophysical Research Letters*, 28(18), 3617–3620. https://doi.org/10.1029/2001GL013552
- Lau, W. K. M., & Kim, K.-M. (2010). Fingerprinting the impacts of aerosols on long-term trends of the Indian summer monsoon regional rainfall. *Geophysical Research Letters*, 37, L16705. https://doi.org/10.1029/2010GL043255
- Lee, L. A., Reddington, C. L., & Carslaw, K. S. (2016). On the relationship between aerosol model uncertainty and radiative forcing uncertainty. *Proceedings National Academy of Sciences of the United States of America*, 113(21), 5820–5827. https://doi.org/10.1073/pnas.1507050113
- Li, X., Ting, M., Li, C., & Henderson, N. (2015). Mechanisms of Asian summer monsoon changes in response to anthropogenic forcing in CMIP5 models. *Journal of Climate*, 28(10), 4107–4125. https://doi.org/10.1175/JCLI-D-14-00559.1
- Malavelle, F. F., Haywood, J. M., Jones, A., Gettelman, A., Larisse, L. C., Bauduin, S., et al. (2017). Strong constraints on aerosol-cloud interactions from volcanic eruptions. *Nature*, 546(7659), 485–491. https://doi.org/10.1038/nature22974
- Marsham, J. H., Dixon, N. S., Garcia-Carreras, L., Lister, G. M. S., Parker, D. J., Knippertz, P., & Birch, C. E. (2013). The role of moist convection in the West African monsoon system: Insights from continental-scale convection-permitting simulations. *Geophysical Research Letters*, 40(9), 1843–1849. https://doi.org/10.1002/grl.50347
- Mellado, J. P. (2017). Cloud-top entrainment in stratocumulus clouds. Annual Review of Fluid Mechanics, 49, 145–169. https://doi.org/10.1146/annurev-fluid-010816-060231
- Michibata, T., Suzuki, K., Sato, Y., & Takemura, T. (2016). The source of discrepancies in aerosol-cloud-precipitation interactions between GCM and a-train retrievals. Atmospheric Chemistry and Physics, 16(23), 15413–15424. https://doi.org/10.5194/acp-16-15413-2016

MÜLMENSTÄDT AND WILCOX 5 of 7



- Mülmenstädt, J., & Feingold, G. (2018). The radiative forcing of aerosol-cloud interactions in liquid clouds: Wrestling and embracing uncertainty. *Current Climate Change Reports*, 4, 23–40. https://doi.org/10.1007/s40641-018-0089-y
- Mülmenstädt, J., Nam, C., Salzmann, M., Kretzschmar, J., L'Ecuyer, T. S., Lohmann, U., et al. (2020). Reducing the aerosol forcing uncertainty using observational constraints on warm rain processes. *Science Advances*, 6(22), eaaz6433. https://doi.org/10.1126/sciadv.aaz6433
- Mülmenstädt, J., Salzmann, M., Kay, J. E., Zelinka, M. D., Ma, P.-L., Nam, C., et al. (2021). An underestimated negative cloud feedback from cloud lifetime changes. *Nature Climate Change*, 11(6), 508–513. https://doi.org/10.1038/s41558-021-01038-1
- Myhre, G., Shindell, D., Bréon, F.-M., Collins, W., Fuglestvedt, J., Huang, J., & Young, P. (2014). Anthropogenic and natural radiative forcing. In Climate change 2013: The physical science basis. Contribution of working group I to the fifth assessment report of the intergovernmental panel on climate change (pp. 659–740). Cambridge, United Kingdom and New York, NY: Cambridge University Press. https://doi.org/10.1017/CBO9781107415324.018
- Oouchi, K., Noda, A. T., Satoh, M., Wang, B., Xie, S.-P., Takahashi, H. G., & Yasunari, T. (2009). Asian summer monsoon simulated by a global cloud-system-resolving model: Diurnal to intra-seasonal variability. *Geophysical Research Letters*, 36, L11815. https://doi.org/10.1029/2009GL038271
- Oreskes, N., Shrader-Frechette, K., & Belitz, K. (1994). Verification, validation, and confirmation of numerical-models in the earth sciences. Science, 263(5147), 641–646. https://doi.org/10.1126/science.263.5147.641
- Parishani, H., Pritchard, M. S., Bretherton, C. S., Terai, C. R., Wyant, M. C., Khairoutdinov, M., & Singh, B. (2018). Insensitivity of the cloud response to surface warming under radical changes to boundary layer turbulence and cloud microphysics: Results from the ultraparameterized cam. *Journal of Advances in Modeling Earth Systems*, 10(12), 3139–3158. https://doi.org/10.1029/2018MS001409
- Parishani, H., Pritchard, M. S., Bretherton, C. S., Wyant, M. C., & Khairoutdinov, M. (2017). Toward low-cloud-permitting cloud superparameterization with explicit boundary layer turbulence. *Journal of Advances in Modeling Earth Systems*, 9(3), 1542–1571. https://doi.org/10.1002/2017MS000968
- Pincus, R., & Baker, M. (1994). Effect of precipitation on the albedo susceptibility of clouds in the marine boundary-layer. *Nature*, 372(6503), 250–252. https://doi.org/10.1038/372250a0
- Polson, D., Bollasina, M., Hegerl, G. C., & Wilcox, L. J. (2014). Decreased monsoon precipitation in the northern hemisphere due to anthropogenic aerosols. *Geophysical Research Letters*, 41(16), 6023–6029. https://doi.org/10.1002/2014GL060811
- Quaas, J., Arola, A., Cairns, B., Christensen, M., Deneke, H., Ekman, A. M. L., et al. (2020). Constraining the Twomey effect from satellite observations: Issues and perspectives. Atmospheric Chemistry and Physics, 20(23), 15079–15099. https://doi.org/10.5194/acp-20-15079-2020
- Ramu, D. A., Sabeerali, C. T., Chattopadhyay, R., Rao, D. N., George, G., Dhakate, A. R., et al. (2016). Indian summer monsoon rainfall simulation and prediction skill in the CFSV2 coupled model: Impact of atmospheric horizontal resolution. *Journal of Geophysical Research*, 121(5), 2205–2221. https://doi.org/10.1002/2015JD024629
- Randall, D., Khairoutdinov, M., Arakawa, A., & Grabowski, W. (2003). Breaking the cloud parameterization deadlock. *Bulletin of the American Meteorological Society*, 84(11), 1547–1564. https://doi.org/10.1175/BAMS-84-11-1547
- Regayre, L. A., Johnson, J. S., Yoshioka, M., Pringle, K. J., Sexton, D. M. H., Booth, B. B. B., et al. (2018). Aerosol and physical atmosphere model parameters are both important sources of uncertainty in aerosol erf. *Atmospheric Chemistry and Physics*, 18(13), 9975–10006. https://doi.org/10.5194/acp-18-9975-2018
- Ridley, H. E., Asmerom, Y., Baldini, J. U. L., Breitenbach, S. F. M., Aquino, V. V., Prufer, K. M., et al. (2015). Aerosol forcing of the position of the intertropical convergence zone since ad 1550. *Nature Geoscience*, 8(3), 195–200. https://doi.org/10.1038/NGEO2353
- Salzmann, M., Ming, Y., Golaz, J.-C., Ginoux, P. A., Morrison, H., Gettelman, A., et al. (2010). Two-moment bulk stratiform cloud microphysics in the GFDL AM3 GCM: Description, evaluation, and sensitivity tests. Atmospheric Chemistry and Physics, 10(16), 8037–8064. https://doi.org/10.5194/acp-10-8037-2010
- Sato, Y., Goto, D., Michibata, T., Suzuki, K., Takemura, T., Tomita, H., & Nakajima, T. (2018). Aerosol effects on cloud water amounts were successfully simulated by a global cloud-system resolving model. *Nature Communications*, 9, 985. https://doi.org/10.1038/s41467-018-03379-6
- Siebesma, A. P., Soares, P. M. M., & Teixeira, J. (2007). A combined eddy-diffusivity mass-flux approach for the convective boundary layer. Journal of the Atmospheric Sciences, 64(4), 1230–1248. https://doi.org/10.1175/JAS3888.1
- Song, F., & Zhang, G. J. (2020). The impacts of horizontal resolution on the seasonally dependent biases of the Northeastern Pacific ITCZ in coupled climate models. *Journal of Climate*, 33(3), 941–957. https://doi.org/10.1175/JCLI-D-19-0399.1
- Song, F., Zhou, T., & Qian, Y. (2014). Responses of East Asian summer monsoon to natural and anthropogenic forcings in the 17 latest CMIP5 models. *Geophysical Research Letters*, 41(2), 596–603. https://doi.org/10.1002/2013GL058705
- Squires, P. (1958). The microstructure and colloidal stability of warm clouds. 1. The relation between structure and stability. *Tellus*, 10(2), 256–261. https://doi.org/10.1111/i.2153-3490.1958.tb02011.x
- Tan, Z., Kaul, C. M., Pressel, K. G., Cohen, Y., Schneider, T., & Teixeira, J. (2018). An extended eddy-diffusivity mass-flux scheme for unified representation of subgrid-scale turbulence and convection. *Journal of Advances in Modeling Earth Systems*, 10(3), 770–800. https://doi.org/10.1002/2017MS001162
- Terai, C. R., Pritchard, M. S., Blossey, P., & Bretherton, C. S. (2020). The impact of resolving subkilometer processes on aerosol-cloud interactions of low-level clouds in global model simulations. *Journal of Advances in Modeling Earth Systems*, 12(11), e2020MS002274. https://doi.org/10.1029/2020MS002274
- Toll, V., Christensen, M., Quaas, J., & Bellouin, N. (2019). Weak average liquid-cloud-water response to anthropogenic aerosols. *Nature*, 572(7767), 51–55. https://doi.org/10.1038/s41586-019-1423-9
- Tsushima, Y., Ringer, M. A., Martin, G. M., Rostron, J. W., & Sexton, D. M. H. (2020). Investigating physical constraints on climate feedbacks using a perturbed parameter ensemble. Climate Dynamics, 55(5–6), 1159–1185. https://doi.org/10.1007/s00382-020-05318-y
- Twomey, S. (1977). Influence of pollution on shortwave albedo of clouds. *Journal of the Atmospheric Sciences*, 34(7), 1149–1152. https://doi.org/10.1175/1520-0469(1977)034<1149:TIOPOT>2.0.CO;2
- Wang, C., Zhang, L., Lee, S.-K., Wu, L., & Mechoso, C. R. (2014). A global perspective on CMIP5 climate model biases. Nature Climate Change, 4(3), 201–205. https://doi.org/10.1038/NCLIMATE2118
- Wang, M., Larson, V. E., Ghan, S., Ovchinnikov, M., Schanen, D. P., Xiao, H., et al. (2015). A multiscale modeling framework model (superparameterized CAM5) with a higher-order turbulence closure: Model description and low-cloud simulations. *Journal of Advances in Modeling Earth Systems*, 7(2), 484–509. https://doi.org/10.1002/2014MS000375
- Wilcox, L. J., Dong, B., Sutton, R. T., & Highwood, E. J. (2015). The 2014 hot, dry summer in northeast Asia. Bulletin of the American Meteorological Society, 96(12), S105–S110. https://doi.org/10.1175/BAMS-D-15-00123.1

MÜLMENSTÄDT AND WILCOX 6 of 7







- Wilcox, L. J., Liu, Z., Samset, B. H., Hawkins, E., Lund, M. T., Nordling, K., et al. (2020). Accelerated increases in global and Asian summer monsoon precipitation from future aerosol reductions. *Atmospheric Chemistry and Physics*, 20(20), 11955–11977. https://doi.org/10.5194/acp-20-11955-2020
- Willetts, P. D., Marsham, J. H., Birch, C. E., Parker, D. J., Webster, S., & Petch, J. (2017). Moist convection and its upscale effects in simulations of the Indian monsoon with explicit and parametrized convection. *Quarterly Journal of the Royal Meteorological Society*, 143(703), 1073–1085. https://doi.org/10.1002/qi.2991
- Zhou, C., & Penner, J. E. (2017). Why do general circulation models overestimate the aerosol cloud lifetime effect?: A case study comparing CAM5 and a CRM. *Atmospheric Chemistry and Physics*, 17(1), 21–29. https://doi.org/10.5194/acp-17-21-2017
- Zhou, S., Huang, G., & Huang, P. (2020). A bias-corrected projection for the changes in East Asian summer monsoon rainfall under global warming. Climate Dynamics, 54(1–2), 1–16. https://doi.org/10.1007/s00382-019-04980-1

MÜLMENSTÄDT AND WILCOX 7 of 7